Author's response to reviews

Title: The Nigerian national blindness and visual impairment survey: rationale, objectives and detailed methodology

Authors:

Brendan Dineen (brendan.dineen@nuigalway.ie)
Clare E Gilbert (Clare.Gilbert@lshtm.ac.uk)
Mansur M Rabiu (mrabiu@hotmail.com)
Fatima Kyari (fatimaygk@yahoo.co.uk)
Tafida Abubakar (drtafida@yahoo.com)
Abdull M Mahdi (abdullmm@yahoo.com)
Christian Ezelum (ccezelum@yahoo.com)
Gabriel Entekume (entekume_66@yahoo.com)
Elizabeth Elhassan (elizabethelhassan@yahoo.co.uk)
Adenike Abiose (adenikeabieseo@yahoo.com)
Hannah Faal (hfaal@sightsavers.org)
J Jiya (jiyajy@yahoo.com)
C P Ozemela (cpozems@yahoo.co.uk)
Pak S Lee (pak.lee@ucl.ac.uk)
Murthy V S Gudlavalleti (gvsmurthy2000@yahoo.com)

Version: 3 Date: 22 August 2008

Author's response to reviews:

Dear Editors,

We have revised the manuscript as per the formatting suggestions that were communicated. The revised manuscript is being uploaded again.

The title has been modified so that capital case has been removed. All key words have been removed from the abstract section as suggested. The authors names are being changed. Abbreviations are also being changed as suggested.

Vertical lines are being removed from the table. The figure numbers have been removed.